

Time-evolving radiative feedbacks in the historical period

Article

Published Version

Creative Commons: Attribution 4.0 (CC-BY)

Open Access

Salvi, P., Gregory, J. M. ORCID: https://orcid.org/0000-0003-1296-8644 and Ceppi, P. (2023) Time-evolving radiative feedbacks in the historical period. Journal of Geophysical Research: Atmospheres, 128 (20). e2023JD038984. ISSN 2169-8996 doi: 10.1029/2023JD038984 Available at https://centaur.reading.ac.uk/113301/

It is advisable to refer to the publisher's version if you intend to cite from the work. See <u>Guidance on citing</u>.

To link to this article DOI: http://dx.doi.org/10.1029/2023JD038984

Publisher: American Geophysical Union

All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other copyright holders. Terms and conditions for use of this material are defined in the <u>End User Agreement</u>.

www.reading.ac.uk/centaur

CentAUR



Central Archive at the University of Reading

Reading's research outputs online



JGR Atmospheres

RESEARCH ARTICLE

10.1029/2023JD038984

Key Points:

- The time-evolution of simulated historical radiative feedback is explained by the variation in the relative sizes of different forcings
- This is because forcings differ in the change they cause to global-mean tropospheric stability per degree of global-mean temperature change
- Effective climate sensitivity is markedly lower to volcanic than greenhouse-gas forcing due to efficient cooling of the Pacific warm pool

Correspondence to:

P. Salvi and J. M. Gregory, pietro.salvi@env.ethz.ch; j.m.gregory@reading.ac.uk

Citation:

Salvi, P., Gregory, J. M., & Ceppi, P. (2023). Time-evolving radiative feedbacks in the historical period. *Journal* of Geophysical Research: Atmospheres, 128, e2023JD038984. https://doi. org/10.1029/2023JD038984

Received 30 MAR 2023 Accepted 24 AUG 2023

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any

© 2023. The Authors.

medium, provided the original work is properly cited.

Time-Evolving Radiative Feedbacks in the Historical Period

Pietro Salvi^{1,2} , Jonathan M. Gregory^{1,3}, and Paulo Ceppi²

¹National Centre for Atmospheric Science, University of Reading, Reading, UK, ²Department of Physics & Grantham Institute, Imperial College London, London, UK, ³Met Office Hadley Centre, Exeter, UK

Abstract We investigate the time-dependence of radiative feedback in the historical period (since the late 19th century), by analyzing experiments using coupled atmosphere–ocean climate models with historical greenhouse gas, anthropogenic aerosol, and natural forcings, each separately. We find that radiative feedback depends on forcing agent, primarily through the effect of cloud on shortwave radiation, because the various forcings cause different changes in global-mean tropospheric stability per degree of global-mean temperature change. The large time-variation of historical feedback driven by observed sea surface temperature change alone, with no forcing agents, is also consistent with tropospheric stability change, and differs from the similarly large and significant historical time-variation of feedback that is simulated in response to all forcing agents together. We show that the latter results from the varying relative sizes of individual forcings. We highlight that volcanic forcing is especially important for understanding the time-variation, because it stimulates particularly strong feedbacks that tend to reduce effective climate sensitivity. We relate this to stability changes due to enhanced surface temperature response in the Indo-Pacific warm pool.

Plain Language Summary Historical climate change, driven by energy flow into the Earth system, is due to a variety of different influences, called "forcing agents." The main ones are greenhouse gases and small particles in the atmosphere (called "aerosols") emitted by human activities, as well as aerosols that are produced in the upper atmosphere by explosive volcanic eruptions. Each of these historical forcing agents varies significantly over the historical period. By studying them separately, we explain how they together explain the evolution of global temperature since the mid-19th century. Differences among the change in global temperature produced by the various forcing agents are explained by their different effects on the vertical contrast of temperature change within the lower atmosphere, because this in turn affects the amount of low cloud, which reflects sunlight and tends to cool the planet. Of all the forcing agents, volcanic aerosols produce the least global temperature change per unit of forcing. This might be because the volcanic aerosol is present for only a few years at a time in the atmosphere, following each eruption.

1. Introduction

In order to understand global-mean surface temperature change T in a changing climate, we use the concept of the Earth energy budget

$$N = F - R = F - \alpha T \tag{1}$$

(Gregory et al., 2002), where *N* is the global-mean radiative imbalance, i.e. the difference between incoming and outgoing radiative flux at the top-of-atmosphere, *F* the effective radiative forcing, and α the radiative feedback parameter. The term $R \equiv \alpha T$ is the *T*-mediated radiative response of the climate system, which occurs so as to oppose the global energy imbalance. It is surprising a priori that a constant α could adequately describe the combination of non-linear feedbacks in the climate system, but much previous work has found this to be a good first approximation.

However, analysis of experiments with general circulation models (GCMs) suggests that α is not truly constant, but time-dependent, both during historical climate change (Andrews et al., 2018, 2022; Dong et al., 2020; Gregory & Andrews, 2016; Gregory et al., 2020) and climate change produced by assumed forcing scenarios, even in cases of fixed forcings such as quadrupled CO₂ (Andrews et al., 2015; Ceppi et al., 2017; Proistosescu & Huybers, 2017; Rugenstein & Armour, 2021; Rugenstein et al., 2019). This time-dependence is largely attributed to the sea surface temperature (SST) "pattern effect" (Stevens et al., 2016) on radiative feedback. It arises because different areas of the Earth's surface warm at different rates (Andrews et al., 2015; Armour et al., 2013;



C. Zhou et al., 2017). The physical mechanism has been linked in previous work to whether warming occurs in more or less convective regions, because this affects tropospheric stability and hence low cloud amount, lapse rate, and water vapor. Increased stability leads to a more negative shortwave (SW) cloud feedback from enhanced low cloud amount and SW reflection. Lapse rate changes also lead to more damping clear-sky feedback through increased longwave (LW) emission to space from the free troposphere, although in a constant relative humidity framework the lapse rate feedbacks should be almost zero in the LW (Jeevanjee et al., 2021). How stability affects radiation through relative humidity changes is less clear, but this effect is simulated to be small (Ceppi & Gregory, 2019).

Part of the historical pattern effect comes from the time-dependence of the combination of the various forcing agents, which cause different patterns and feedbacks, as shown by prior studies (Ceppi & Gregory, 2019; Marvel et al., 2016; Salvi et al., 2022; Shindell, 2014; Smith & Forster, 2021). By comparison with experiments forced by the individual historical forcing agents, we can evaluate their contributions to the evolution of historical feedbacks. Thus, in historical experiments of the Coupled Model Intercomparison Project Phase 5 (CMIP5), Gregory et al. (2020) found that the historical greenhouse gas (GHG) feedback does not vary significantly in time, and that natural forcing affects the time-evolution of net feedback, because the feedback in response to forcing from volcanic stratospheric aerosol is less amplifying than to GHG forcing.

The more recent CMIP6 data set (Phase 6 of CMIP) includes the experiments of the single-forcing Detection and Attribution Model Intercomparison Project (DAMIP; Gillett et al., 2016) with coupled atmosphere-ocean GCMs (AOGCMs), and the Radiative Forcing Model Intercomparison Project (RFMIP; Pincus et al., 2016) with atmosphere-only GCMs (AGCMs). The latter allow model-specific diagnosis of historical effective radiative forcings as a function of time. This provides an advantage over the CMIP5 data set available to Gregory et al. (2020), who assumed the same forcing for all AOGCMs in the absence of further information. Dong et al. (2021) showed that feedbacks in CMIP6 AOGCM historical experiments are mainly driven by GHG forcing from around 1940 onward. In the present work, we use CMIP6 experiments to explain time-dependence of historical feedback in terms of the time-variations of and differences among the forcing agents.

CMIP5 and CMIP6 AOGCMs generally fail to reproduce the observed historical SST pattern of recent decades, especially in the Pacific, and the consequent large positive α (Andrews et al., 2018, 2022; Dong et al., 2019, 2021; Gregory & Andrews, 2016; Gregory et al., 2020; Wills et al., 2022). Since AOGCMs are essential for producing physically based estimates of future warming, we need to know why they do not reproduce the past in this respect. Not having a multi-centennial observational SST record for comparison, we cannot tell whether the relevant SST pattern is an unusual unforced variation, or a response to recent forcing. In either case, improved understanding of simulated historical feedbacks will be useful in explaining the difference from the observed record.

The objectives of this paper are to investigate the evolution of radiative feedbacks in the historical period, and to explain this through the time-varying mix of historical forcing agents. The following sections are organized thus: Section 2 details the methods and data used. Section 3 is a short section confirming the additivity of individual forcings. Section 4 is an analysis of historically relevant forcing agents and their mean associated radiative feedbacks across the historical period. Section 5 discusses how feedbacks for different historical forcing agents evolve over time, whilst Section 6 assesses the extent to which we can explain the evolution of historical feedbacks through the mix of historical forcing agents. Section 7 then discusses possible reasons for the significantly less-amplifying feedbacks from volcanic eruptions, which are relevant to the evolution of historical feedbacks. Finally, Section 8 provides a summary and conclusions of this work, and outlines needed future work.

2. Data and Methods

2.1. Experiments, Models, and Data Processing

The data used in this paper are detailed in Table 1, and were obtained from the Earth System Grid Federation archives, which hold CMIP data from multiple centers (Cinquini et al., 2014). The data include historical single-forcing simulations with both AOGCMs (hist-x, from DAMIP; Gillett et al., 2016) and AGCMs with fixed SSTs (piClim-histx, from RFMIP; Pincus et al., 2016), where x is a short name for the forcing agent. We chose variants with the same initialization ("i1" variant label), physics and forcing definition as in piControl, except where noted in Table 1. We included all realizations that contained all of the required variables, and our results from each experiment of each model are ensemble averages. We calculated multi-model mean (MMM) values from the individual model ensemble averages, with equal weighting for each model.



Model					Realizations			
CanESM5 p2f1	1	1-10	1-10	1–9	1	1	1	1
CESM p1f1	1	×	×	×	×	×	×	1
CNRM-CM6-1 p1f2	1	1-10	1-10	1–9	1	1	1	1
GISS-E2-1-G p1f1 ^b	1	1-5	1-10	1–9	1	1	1	×
HadGEM3-GC31-LL p1f3	1	1-4	1-5	14	1	1–3	1–3	1
IPSL-CM6A-LR p1f1	1	1-10	1-10	1–9	1	1	1	1
MIROC6 p1f1	1	1–3	1-10	1–3	1	1–3	1–3	1
MRI-ESM2-0 p1f1	1	×	×	×	×	×	×	1
NorESM2-LM p1f1	1	1–3	1–3	1–3	1	1–3	1–3	×
	piControl ^a	hist-GHG and hist-aer	Historical	hist-nat	piClim-control	piClim-hist(aer/ghg/all)	piClim-histnat	amip-piForcing
<i>Note.</i> "Realizations" refers to th ^a All niControl runs used the n1	he CMIP6 r variar. 1 labels, excent fo	nt labels, with noted physics (<i>j</i> or CNRM-CM6-1 which used	 and forcing (f) all for GISS 	labels. All var EE2-1-G. the v	iants used the same i ariant chosen for hist	l initialisation label. Unavailab- -nat was B. since fl was unava	ale data are marked by the set of \mathcal{P} does no	y an X. t include solar t

Throughout this work, we omit analysis of individual models and instead focus on MMM results. Results from individual models vary substantially, but investigating issues particular to specific models is outside the scope of this work. In order to minimize regression dilution, we calculate least squares regressions using MMM results, rather than calculating the MMM of regression results (Gregory et al., 2020).

Also included are data from the amip-piForcing experiments which use AGCMs with SSTs prescribed according to the Atmosphere Model Intercomparison Project II data set, derived from historical observations (Andrews et al., 2018, 2022; Gregory & Andrews, 2016; Hurrell et al., 2008; Webb et al., 2017). Only single ensemble members were included for each model for these experiments, but this is not expected to have a substantial impact on the results, because feedback time-variations in amip-piForcing are largely dictated by the prescribed SSTs, since atmospheric unforced variability is comparatively small (Gregory & Andrews, 2016). The set of models with amip-piForcing experiment outputs differs slightly from that used for historical single-forcing experiments (Table 1), but this is unimportant for the results discussed below.

Global averages refer to area-weighted averages of the variables of interest, except for stability change *S* which refers to the average change in EIS (Wood & Bretherton, 2006) over oceans equatorwards of 50°S and 50°N. This choice follows Ceppi and Gregory (2019) who found that area-mean EIS change in this region is well correlated with the global radiative response in observations and in two GCMs.

2.2. Calculating Forcing and Climate Response

We calculate the forcing and the surface-temperature-mediated climate response from the data detailed in Table 1 for each forcing x.

The TOA radiative imbalance, N_{piClimx} , from the fixed-SST experiment piClim-*x* relative to piClim-control equals the effective radiative forcing F_x according to Equation 1, since $R \approx 0$ with SST not allowed to change. The surface forcing is evaluated in a similar way, as the difference in net downward surface heat flux between piClim-*x* and piClim-control, and includes rapid adjustment (like the ERF). Unlike TOA forcing, the surface forcing has sensible and latent heat as well as radiative flux components. The time-mean global averages of the TOA and surface forcings are equal if energy is conserved, since the global heat content of the atmosphere must have zero trend in the steady state of the piClim-*x* experiment, but locally they are not equal.

The TOA radiative imbalance, N_{histx} , from the coupled experiment hist-*x* relative to piControl includes both forcing and response that is, $N_{\text{histx}} = F_x - R_x$. Therefore we subtract N_{histx} from $N_{\text{piClimx}} = F_x$,

$$N_{\text{piClim}x} - N_{\text{hist}x} = F_x - N_{\text{hist}x} = R_x, \qquad (2)$$

to isolate TOA radiative changes R_x that are mediated by SST change due to agent x.

2.3. Volcanic Events

volcanic forcings whereas β does. The difference between 1 and β for hist-nat is that β forcings include a correction to ozone-related heterogeneous reactions.

We identify radiatively significant volcanic forcing events by inspecting the time series of effective forcing in piClim-histnat. We used SciPy's peak finding algorithm, which uses a simple comparison of neighboring values, to find negative peaks of greater than 0.3 W m⁻² magnitude in effective forcing. Identified peaks of negative forcing are shown in red in Figure 1. The first one, which occurs in 1861, lines up roughly with the Dubbi (Wiart & Oppenheimer, 2000) and Makian (Picas & Grab, 2020) eruptions of that year, but is unidentified in the data set of Durack and Taylor (2022) based on ice core data (Arfeuille et al., 2014). We call this one

Table 1



Figure 1. Annual means of effective top-of-atmosphere (TOA) radiative forcing over time (top) and global average surface temperature change (bottom) in hist-nat over the 1850–2015 historical period. The seven largest peaks of negative forcing of the period (corresponding to the seven most climatologically significant volcanic eruptions) are marked by red dots, whilst years of measured negative forcing (i.e., volcanic forcing, which makes up "hist-nat-volc," defined in the text) are shaded in gray. Instantaneous radiative forcing from changes in downwelling shortwave at the TOA is shown in blue.

the "Early" eruption. All the later identified peaks accord with lists of historically significant volcanic eruptions (Mass & Portman, 1989; Stenchikov et al., 1998): Krakatoa (1883), Santa María/Mount Pelée/Soufrière (1902, which we refer to as "Santa María"), Katmai/Novarupta (1912, which we refer to as "Novarupta"), Agung (1963), El Chichón (1982), and Pinatubo (1991). Many other eruptions occurred over the 1850–2015 historical period, including some highly explosive ones (Newhall & Self, 1982) such as Mt. St. Helens and Cerro Hudson (Schmidt et al., 2018), but these have relatively little global average radiative impact.

We regard negative effective forcing in hist-nat (gray bars in Figure 1) as indicating the presence of substantial volcanic forcing. Zero forcing in hist-nat indicates no forcing difference from piControl, which contains a



Figure 2. Sum of flux imbalances (N) for piClim-histall and amip-piForcing experiments against N for the amip-hist experiment. Each dot is a yearly average, for the model mean of models that had data for all required experiments for this figure (CanESM5, IPSL-CM6A-LR, CNRM-CM6-1, and MIROC6). The linear fit gives a gradient *m* of correlation ρ . This linear fit uses orthogonal distance regression, which finds a least squares fit according to error along both axes. This is done by accounting for approximately double the variance in the *y*-axis, which is a sum of two experimental outputs, compared to along the *x*-axis, which is obtained from a singular experimental output. The same amip-hist variants were used as for amip-piForcing.

constant background volcanic aerosol (equivalent to the historical average over 1850–2015; Eyring et al., 2016). Consequently, hist-nat has a weak *positive* forcing relative to piControl during non-volcanic periods (Gregory et al., 2013), such as 1920–1960, when warming occurs in hist-nat (Figure 1b). Solar forcing is also present in hist-nat, but is comparatively small.

3. Confirming That Forcing and Response Combine Linearly

In order to check that historical N is a linear combination of F and R (as assumed by Equation 2), we use the TOA radiative imbalance:

 $N_{\text{amip-hist}} = F - R$ from the amip-hist experiment, which prescribes the historical observed SST and historical forcing agents (T. Zhou et al., 2016b),

 $N_{\text{amip-piForcing}} = -R$ from the amip-piForcing experiment, which prescribes the same SST but no forcing agents so F = 0, and

 $N_{\text{piClim-histall}} = F$ from the piClim-histall experiment, which prescribes historical forcing agents but no SST change, relative to piClim-control (as in Section 2.2).

We expect

$$N_{\text{amip-hist}} = F - R = N_{\text{piClim-histall}} + N_{\text{amip-piForcing}}.$$
 (3)

We test Equation 3 through Figure 2, which shows that this holds over the historical period. An orthogonal distance regression (Boggs & Rogers, 1990)



10.1029/2023JD038984



Figure 3. Multi-model mean responses R against surface temperature change T, for (a) hist-aer, (b) hist-GHG, (c) hist-nat, and (d) the historical experiments. Each dot marks a yearly average, with linear best-fits (of gradient α , with standard regression error) in solid gray.

is performed, using the fact that the variance in the *y*-axis should be approximately double the variance in the *x*-axis since the *y*-axis is the sum of two model integrations with independent unforced variability, as opposed to the single model output used in the *x*-axis. This fit has high correlation ($\rho = 0.965$) and gradient (0.962) close to the 1:1 line. Thus the climate response in AGCMs to the combination of historical forcings can be accurately obtained by subtracting the forcing-only experiment from the experiment involving both forcing and SST change. We therefore assume the same applies for individual forcing agents in AOGCMs, for which there are not corresponding experiments to test it.

4. Dependence of Radiative Feedback on Forcing Agent

For each time-dependent forcing in the MMM of historical coupled experiments, R and T are well-correlated during 1850–2015 (Figure 3), but the best-fit constant α from the regression slope of R against T over the whole period depends on the forcing agent. As in Salvi et al. (2022), historical aerosol forcing is found to cause more amplifying feedbacks (smaller magnitude α) than historical GHG forcing, and historical net forcing causes less amplifying feedback (larger magnitude α) than historical GHG forcing. As argued in Salvi et al. (2022), these two findings are related under the assumption that the net historical forcing is primarily a combination of positive GHG forcing and negative aerosol forcing. This argument is shown geometrically in Figure 4: approximating the historical response as a sum of historical aerosol and GHG forcing alone. Panel (a), gives a gradient of R against T that is larger than the gradient for historical GHG forcing alone. Panel (b) shows that, if the volcanic response is relatively small then it can have little impact on the historical gradient even if its associated α is large.

In order to understand the dependence of α on historical forcing agents, we use the refined model of Ceppi and Gregory (2019). The Earth energy budget model of Section 1 gives





(4)



if α is constant. To account for the variation of α shown by GCM experiments, Ceppi and Gregory (2019) proposed a model wherein *R* is described as a combined response to both *T* and global average atmospheric stability change *S*:

$$R = \tau T + \sigma S \Rightarrow \alpha = \frac{dR}{dT} = \tau + \sigma \frac{dS}{dT},$$
(5)

where

$$\tau = \frac{\partial R}{\partial T} \tag{6}$$

and

$$\sigma = \frac{\partial R}{\partial S}.$$
(7)

Through uniform warming experiments in two models, Ceppi and Gregory (2019) found that τ and σ are approximately constant in each model, *i.e.* the radiative response and stability change are both roughly linear with uniform global surface temperature change. Therefore, in their experiments, differences in $\alpha = dR/dT$ for different forcing agents are explained by differences in dS/dT (Equation 5).

To test whether this explanation applies to CMIP6 historical forcings, we calculate linear regression slopes $\alpha = dR/dT$ and dS/dT from the MMM R(t), S(t), and T(t) of each of the hist-*x* experiments and the historical experiment, using data for 1850–2015, except that for "hist-nat-volc" we use the volcanic forcing years only. There is a positive linear relationship between α and dS/dT (Figure 5a). The relationship is consistent with Equation 5 and with the hypothesis of Ceppi and Gregory (2019). It indicates that MMM feedback differences between historical natural and anthropogenic forcings are due to different stability responses, as previously found for hist-aer and hist-GHG (Salvi et al., 2022).

Furthermore, there is a strong and significant relationship ($\rho > 0.75$) between dS/dT and α across historical forcing agents in every model by itself, except in MIROC6. Although τ and σ (Equation 5) might be different in each model (we have not investigated this statistically), when all models and forcing agents are considered together the relationship between dS/dT and α remains significant ($\rho = 0.72$, p < 0.01 of the null hypothesis of no relationship, Figure 6). Therefore from both the individual models, as well as the MMM, we conclude that the differences among the forcing agents in feedback for the full historical period can be explained by the different changes in stability that they produce.

Taking the difference between Equation 5 for forcing agent x and GHG gives

$$\alpha_x - \alpha_{GHG} = \sigma(dS/dT_x - dS/dT_{GHG}) \tag{8}$$

if τ and σ are the same for all forcing agents in the MMM, as found for individual models by Ceppi and Gregory (2019). Equation 8 is the equation of a straight line through the origin with slope σ . The regression line of $\alpha_x - \alpha_{GHG}$ against $dS/dT_x - dS/dT_{GHG}$ (Figure 5a) has this form. Hence we can identify σ for the MMM with the slope of the regression line, which is 5.05 ± 0.7 Wm⁻²K⁻¹, and is consistent, within uncertainty, with σ for both models analyzed by Ceppi and Gregory (2019).

To gain further physical insight, we break α down into cloud and non-cloud feedbacks *i.e.* those from water vapor, lapse rate, and albedo changes. For SW components, we use the "approximate partial radiative perturbation" method (APRP, Taylor et al., 2007), and for LW the radiative kernel method (Pendergrass et al., 2018; Soden et al., 2008). The kernel method can also be used for SW and is expected to agree with APRP (Zelinka et al., 2020), but we find that the APRP method gives a smaller residual and so is preferred.

Net non-cloud feedbacks (Figure 5b), mostly in the LW (Figure 5h), show some contribution to overall differences in α between forcing agents. However, SW cloud feedbacks (Figure 5f) are the most important, with both the highest spread in and correlation between α and dS/dT. This is in agreement with the findings from Ceppi and Gregory (2019), and consistent with the dependence of low cloud amount on stability (Andrews & Webb, 2018; Ceppi et al., 2017; Ceppi & Gregory, 2019; C. Zhou et al., 2016a).





Figure 5. Multi-model mean radiative feedbacks α against dS/dT, with values calculated from regression over the full historical period (1850–2015) for hist-GHG, hist-aer, hist-nat, and historical. Also shown are values for regression over only volcanic years of hist-nat ("hist-nat-volc," see Figure 1 for selection). We show here a breakdown into non-cloud and cloud feedback components, in SW (from the APRP method) and LW (from radiative kernels). Values are shown relative to the hist-GHG experiment, for easier comparison between non-cloud and cloud contributions to combined feedbacks. Axes are shared throughout. Gradients *m* of least squares fits are shown in each panel. Error bars denote estimated regression errors.









Figure 7. Forcing over time in the historical case and for simulations of component historical forcing agents. Values were calculated from RFMIP piClim-hist*x* experiments with piClim-control as the control. Vertical gray lines demarcate the three different periods of different forcing flavors.

5. Time-Evolution of Radiative Feedback

Having looked at radiative feedbacks calculated using regressions across the full historical period (1850–2015), we next assess how historical feedbacks change with time during this period. Figure 7 shows the ERF of each forcing agent as a function of time. The historical period can be roughly divided into three: an early period where both anthropogenic aerosol (AA) and GHG forcings are small but several significant eruptions occur (1850–1914); a middle period where both AA and GHG forcing are substantial but no major volcanic eruptions occur (1915–1959); and a late period which contains strong forcing from both AAs and GHGs alongside multiple major volcanic eruptions (1960–2015).

Natural forcing exhibits a highly variable temporal structure (*pink* in Figure 7), dominated by the contribution of volcanic forcings, while solar forcing remains weak throughout (Figure 1a). By contrast to the smoothly evolving hist-GHG and hist-aer forcings, natural forcing varies over very brief timescales as forcing from volcanic eruptions decays rapidly over a few years and many years can pass between significant volcanic eruptions (Ammann et al., 2003).

Figure 8 shows the corresponding time-evolving feedbacks and dS/dT for regressions over 30-year sliding windows, following the approach of Gregory and Andrews (2016) and Gregory et al. (2020), compared with the feedbacks from regression over the full time series (dashed horizontal lines). Both α and dS/dT vary significantly over time in all cases except hist-GHG. Furthermore, except for hist-GHG and excluding the middle period in hist-nat (because it contains no significant forcing), there is a high correlation ($\rho > 0.75$) between variations in time of α and dS/dT for each forcing.

However, the significance of correlations between $\alpha(t)$ and dS/dT(t) cannot be determined from the overlapping 30-year regression windows due to autocorrelation, which is difficult to allow for. Hence, we examine the relationship between $\alpha(t)$ and dS/dT(t) using non-overlapping windows. We divide the 1865–2015 period into five 30-year windows for hist-aer, hist-GHG, and the historical cases. The amip-piForcing simulation only has four windows, due to a later start date. We also only use four windows for the hist-nat case as we ignore the 1925–1955 window, which is free from significant volcanic eruptions.

The bottom row of Figure 8 (panels k–o) shows linear fits of $\alpha(t)$ and dS/dT(t) for all sliding windows (dots, dashed gray line) and using only non-overlapping windows (black circles with error bars, solid black line). From panels k and l, the different best-fits when using overlapping or non-overlapping windows suggest that there is some over-fitting when using overlapping windows. Using non-overlapping windows, the correlation in time of $\alpha(t)$ and dS/dT(t) is high ($\rho > 0.8$) in all cases except for hist-GHG forcing, and the regression slopes of $\alpha(t)$ against dS/dT(t) are significantly greater than zero (p < 0.05) for hist-aer and hist-all. For hist-nat and amip-piForcing they might not be significant but are nonetheless unlikely to arise from random noise (p < 0.11 and p < 0.06, respectively). Because the four experiments are independent, we can multiply together their *p*-values to obtain the probability of obtaining a positive correlation in all four cases if there were actually no relationship between α and dS/dT in models. The product is very small ($p \simeq 10^{-5}$). We interpret this as strong evidence of the existence of a relationship.

In the following subsections, we go into more detail for each of the forcings.

5.1. Hist-aer

Historical AA forcing (Figure 8a) causes a relatively large α between around 1920–1930, in contrast to the lower α (compared to under GHG forcing) that was calculated over the full historical period (Figure 5). This feature is robust in *dS/dT* across models (not shown) but not obviously related to the evolving patterns of aerosol forcing, which are known to have shifted from being mainly over Europe and north America to their increasing prominence in east and south Asia (Lamarque et al., 2010; Wang & Wen, 2021), despite shifting forcing patterns being tied to different efficacies (Persad & Caldeira, 2018). The reason for the change in feedbacks for hist-aer could be related to changing relative concentrations of different AAs, as highlighted by an increase in radiative forcing from sulphates relative to black carbon over the historical period (Myhre et al., 2014, their Figure 8.8). A shift



10.1029/2023JD038984



Figure 8. Top and middle rows: Time-evolving feedback parameter α (top row) and dS/dT (middle row) for various historical experiments as calculated using 30-year sliding windows. Colored shaded regions denote regression errors. Dashed time-series in the top row show best linear fits of dS/dT to α using the non-overlapping windows from 1855 to 2015 in all experiments, except hist-nat where the fit omits the 1915–1945 window during which no significant volcanic forcing occurs (see Figure 1). Horizontal dashed lines show regressions across the full time series (as in Figure 5a). Pearson correlation coefficients of these fits to α (ρ), as well as regression p-values, are shown. In brackets, we show ρ and p for fits using overlapping windows. For hist-nat, again, windows over the period free from eruptions are ignored in the fit. Peaks of volcanic forcing are shown by red crosses for the hist-nat and historical cases. Vertical gray lines demarcate the three periods of different forcing flavors. Bottom row: year-by-year scatter of α and dS/dT from 30-year windows. Circles with black edges show non-overlapping windows, whilst dots show windows that overlap with these. Linear fit coefficients for the gradient σ and intercept τ are shown, across independent windows (black, solid line fit) and all available windows (gray, dotted line fit). The window free of significant volcanic eruptions is ignored for hist-nat as above, and is shown by a white circle with a black edge in panel (m).

from high α when black carbon dominates (around 1925) to low α as sulphates become more important (1950 onwards) would be consistent with the finding by Ceppi and Gregory (2019) of larger α (lower sensitivity) from black carbon than sulphate forcing. This is an uncertain explanation, however, as other works found opposite results (Nordling et al., 2021; Richardson et al., 2019). Furthermore, feedbacks remain closer to the value from regression across the full period *after* forcing becomes more significant (*i.e.*, after the year 1900), which suggests that this could be the result of internal variability.

5.2. Hist-GHG

For hist-GHG (Figures 8b and 8g), the low correlation ($\rho = -0.44$) across time between dS/dT and α is due to the lack of forced variations in these two parameters. Feedbacks vary closely around the value obtained from regression across the full period. With the hist-GHG forcing pattern remaining homogeneous throughout the historical period, only its magnitude slowly increasing, the inter-annual evolution of dS/dT and α is dominated by unforced variability, unlike the other cases where the forcing patterns and/or magnitudes vary more rapidly in time. This is highlighted by the good correlation of R with T over a larger T range than for the other experiments (Figure 3), such that for hist-GHG both dR/dT and dS/dT are nearly constant, and their correlation is



negligible in time. The lack of dS/dT and α variability in hist-GHG can be seen clearly from the tight scatter of points compared to the greater spread seen in other cases (Figure 81 compared with other panels of the bottom row).

5.3. Hist-nat

The hist-nat case is dominated by volcanic forcing (Figures 8c and 8h). This causes substantial variation in feedbacks across the period. Low correlation between dS/dT and α in the middle period lines up with a lack of significant eruptions, which means there is no forced signal then. In other periods, the hist-nat experiment shows significant variations in α and dS/dT which are possibly related to different eruptions causing different feedbacks, for example, via differences in the spatial distributions of stratospheric aerosol and hence radiative forcing. Strong volcanic forcing periods tend to have higher α than for hist-GHG forcing, which is consistent with larger alpha when considering the entire period.

5.4. Historical

In the historical all-forcing case (Figures 8d and 8i) before 1900, variations in α and dS/dT very closely mirror those found in hist-nat, because AA and GHG forcing are relatively small. After 1980, α tends toward similar values as in hist-GHG. This could be due to a combination of factors, namely that hist-GHG forcing dominates toward the end of this historical period, whilst the strong Pinatubo eruption, with higher α , and aerosol forcing, with lower α , likely cancel out in their effects on radiative feedbacks.

The reason for the large variations in α and dS/dT in the middle period in the historical all-forcing case is less clear. One possible explanation is that AA and GHG forcing tendencies roughly cancel out (Figure 7), such that regressions in temperature over this period are sensitive to unforced variations in *T*. Alternatively, variations in α for hist-aer may explain it as follows: high α from hist-aer during 1910–1930 results in α being *less* positive in the historical all-forcing case than for hist-GHG over the same period, whilst low α for hist-aer during 1935–1955 results in *more* positive α for the historical all-forcing case (refer to Figure 4).

5.5. Amip-piForcing

The amip-piForcing experiment is different from the DAMIP historical experiments, since it prescribes observed historical SSTs instead of prescribing forcings and allowing SSTs to evolve accordingly. The observed SST pattern imposed in amip-piForcing differs substantially from that simulated by coupled models under historical forcing (Wills et al., 2022). Consistent with this, Figures 8e and 8j shows a different time-evolution of radiative feedbacks in amip-piForcing to the historical all-forcing case. While we do not expect the coupled MMM to reproduce the variability found in the observed single realisation of historical climate, the disagreement may still reflect a failure of models to recreate historical SST patterns, whether forced or unforced (Dong et al., 2021; Gregory et al., 2020; Wills et al., 2022).

Given that volcanic forcing seems to have strong effects on the evolution of feedbacks in the hist-nat and historical cases, we compare the evolution in hist-nat to amip-piForcing. In both, α is high at the start (but the earliest period is missing in amip-piForcing, limiting the comparison), decreases in the early 20th century, and rises steadily toward the end of the historical period. The timing of these features does not line up well, however. This could point to a couple of issues. Considering that volcanic eruptions are highly variable in the forcing and response that they produce among individual ensemble members of historical and hist-nat simulations, historical simulations might be unable to capture the actual historical volcanic response because of unforced variability affecting the state of the atmosphere at the time of the eruption. Otherwise, there may be inaccuracy in the prescribed volcanic forcing in the historical simulations; this particularly applies to the very strong Krakatoa eruption, since this occurred well before the satellite record.

Observational errors in the AMIP-II SST boundary condition data set may also play a role (Andrews et al., 2022; Fueglistaler & Silvers, 2021).



10.1029/2023JD038984



Figure 9. Model results from the historical case (*black*) compared to the sum of model results from hist-nat, hist-GHG, and hist-aer cases (*blue*), along with the reconstruction using fixed α for anthropogenic aerosol and GHG forcings but time-varying α for natural forcing (*red*, detailed in the text). (a–c) 3-year running means of (a) the radiative response, (b) 50S-50N EIS changes over ocean, (c) surface air temperature change. (d–f) 30-year regressions of same variables as to their left (d–e) over surface temperature and (f) over time. Note that for *T* and *dT/dt*, the reconstruction and sum are identical as the reconstruction uses the sum of model output temperatures.

5.6. Dependence of σ on Forcing Agent

From the linear regression to time-variation in non-overlapping windows, we obtain slopes σ and intercepts τ for each forcing agent (Equation 5). Values differ, but are not generally statistically incompatible between pairs of forcing agents. It is relevant to note that the smallest standard errors are estimated in the historical case, which involves multiple forcing agents. While the relative importance of forcings varies in time, the historical $\alpha(t)$ can be explained well with constant τ and σ , suggesting that these parameters do not depend strongly on the forcing agent. The value here of $\sigma = 3.88 \pm 0.18$ is not significantly different to the $\sigma = 5.05 \pm 0.7$ obtained in Section 4.

6. Time-Dependence of Feedback Related to Variations in Forcing

To quantify contributions of different forcing agents, which drive different stability changes and thus different feedbacks, to historical radiative feedbacks, we reconstruct the historical response $R_{rec}(t)$ from the results of historical single-forcing experiments through

$$R_{\rm rec}(t) = \alpha_{\rm GHG} T_{\rm GHG}(t) + \alpha_{\rm AA} T_{\rm AA}(t) + R_{\rm nat}(t)$$
(9)

that is, we sum the temperature changes predicted by the hist-aer and hist-GHG experiments multiplied by fixed feedback parameters calculated from the full 1850–2015 period (the same values as in Figure 3), and add in the response contributions from the hist-nat $R_{nat}(t)$ case directly. We follow a similar procedure to reconstruct stability change

$$S_{\rm rec}(t) = \left(\frac{dS}{dT}\right)_{\rm GHG} T_{\rm GHG}(t) + \left(\frac{dS}{dT}\right)_{\rm AA} T_{\rm AA}(t) + S_{\rm nat}(t).$$
(10)

using the same constant $(dS/dT)_{GHG} = 0.125$ and $(dS/dT)_{aer} = 0.072$ as shown by the horizontal dashed lines in the middle row of Figure 8).

Fixed α_{GHG} and α_{AA} are justified since there is little variability in $\alpha_{GHG}(t)$ over time, and much of the variability in $\alpha_{AA}(t)$ occurs earlier on when the AA forcing is relatively weak; later on, variations are relatively small as aerosol forcing becomes stronger. Hence, these variations should not contribute strongly to the historical evolution of feedbacks since the early period is dominated by volcanic forcing. α_{nat} , however, varies significantly in time such that fixed α_{nat} does not accurately apply. Finally, we reconstruct temperature change $T_{rec}(t)$ through



$$T_{\rm rec}(t) = T_{\rm GHG}(t) + T_{\rm AA}(t) + T_{\rm nat}(t).$$
 (11)

The reconstructed historical R_{rec} and S_{rec} are shown in Figures 9a and 9b in *red*. Figures 9a–9c show the sums R_{sum} , S_{sum} and T_{sum} from the hist-*x* experiments in *blue* (as 3-year running means, to reduce short-term unforced variability):

$$R_{\text{sum}}(t) = R_{\text{GHG}}(t) + R_{\text{AA}}(t) + R_{\text{nat}}(t), \qquad (12)$$

$$S_{\text{sum}}(t) = S_{\text{GHG}}(t) + S_{\text{AA}}(t) + S_{\text{nat}}(t), \qquad (13)$$

$$T_{\rm sum}(t) = T_{\rm rec}(t). \tag{14}$$

The CMIP6 MMM historical R_{hist} , S_{hist} and T_{hist} are shown in *black*.

The left column of Figures 9a–9c shows that the reconstruction explains R_{hist} and S_{hist} well, with correlations of $\rho(R_{\text{rec}}, R_{\text{hist}}) = 0.90$ and $\rho(S_{\text{rec}}, S_{\text{hist}}) = 0.79$. The correlation is not expected to be perfect because of unforced variability, missing forcings in the reconstruction (ozone, land-use change) and non-linearities in responses to different forcings, though the high correlations suggest these factors are minor. Nonetheless, these are higher correlations than $\rho(R_{\text{sum}}, R_{\text{hist}}) = 0.83$ and $\rho(S_{\text{sum}}, S_{\text{hist}}) = 0.67$. This may be due to summed unforced variability in the sum of responses, with greater unforced variability in R and T.

The time-dependent $\alpha = dR/dT$ and dS/dT (Figures 9d and 9e) are calculated in three ways:

$$\alpha_{\rm hist} = \frac{dR_{\rm hist}}{dT_{\rm hist}} \qquad \qquad \alpha_{\rm rec} = \frac{dR_{\rm rec}}{dT_{\rm sum}} \qquad \qquad \alpha_{\rm sum} = \frac{dR_{\rm sum}}{dT_{\rm sum}}, \tag{15}$$

$$\frac{dS}{dT}\Big|_{\text{hist}} = \frac{dS_{\text{hist}}}{dT_{\text{hist}}} \qquad \qquad \frac{dS}{dT}\Big|_{\text{rec}} = \frac{dS_{\text{rec}}}{dT_{\text{sum}}} \qquad \qquad \frac{dS}{dT}\Big|_{\text{sum}} = \frac{dS_{\text{sum}}}{dT_{\text{sum}}},$$
(16)

in each case by regression in a 30-year sliding window. For α and dS/dT, the first estimate represents the actual time-variation; the second estimate tests the reconstruction assuming α constant in time but different for GHG and AA; and the third estimate tests the additivity of the individual responses. Note that α_{hist} is the same here (Figure 9d) as in Figure 8d.

The top-right panels of Figures 9d and 9e shows that α_{hist} and dS/dT_{hist} for the CMIP6 MMMs are quite well reproduced by both the reconstruction and the sum, with statistically significant correlations (p < 0.01 of zero regression slope): $\rho(\alpha_{rec}, \alpha_{hist}) = 0.53$, $\rho(dS/dT_{rec}, dS/dT_{hist}) = 0.62$, $\rho(\alpha_{sum}, \alpha_{hist}) = 0.56$, $\rho(dS/dT_{sum}, dS/dT_{hist}) = 0.72$. In the early period, α_{rec} and dS/dT_{rec} are closer than α_{sum} and dS/dT_{sum} to α_{hist} and dS/dT_{hist} . This may point to this period containing significant unforced variability, reflected in the sum of responses but less present in the reconstruction as unforced variability in T is smaller than in R. Around 1940, the reconstruction and the sum both give underestimates, especially for α . These periods also have small trends in T (Figure 9f), which suggests that the discrepancies could be due to low signal-to-noise. Nonetheless, the reconstruction captures the shape of the time-variation of α_{hist} and dS/dT_{hist} . This supports the reasoning that AOGCM-derived historical feedbacks can be interpreted in terms of a time-varying mix of GHG, AA and volcanic forcings. Hence, any non-linearities that arise from the combination of these agents do not play a first-order role for the time-variation in historical feedbacks.

7. Explaining the Lower Climate Sensitivity of Volcanic Forcing

As highlighted by Figure 5, the overall effect of volcanic forcing is to produce significantly less amplifying radiative feedback (lower climate sensitivity) than GHGs. This can be seen when looking at α for hist-nat (all years, *pink*) and is emphasized by regressions over only volcanic forcing years (hist-nat-volc, *red*). In order to understand this, we restrict our attention to the two largest volcanic eruptions of this period: Krakatoa (1883) and Pinatubo (1991). These eruptions have a clear signal in all of the models analyzed here, with correspondingly the largest responses in the MMM (Figure 1).

In the analysis that follows, we consider 1882–1886 (for Krakatoa) and 1990–1994 (for Pinatubo), *i.e.* for each of these eruptions we have taken the year of the eruption, and the three subsequent years. These years following the peak forcing are included to allow the volcanic forcing signal to decay to a level which cannot be



10.1029/2023JD038984



Figure 10. (a–c) Zonal patterns of time-mean historical (a) forcing at the TOA, (b) surface forcing, and (c) surface temperature change, each divided by its global average, during 1995–2015 from hist-aer, hist-ghg and all historical agents, and during the years with significant forcing from the two largest volcanic events in the historical period (Krakatoa and Pinatubo). (d–f) Distributions of TOA effective forcing (d), surface forcing (e) and surface temperature change (f) over the sea within the tropics, as functions of climatological sea surface temperature, time mean over the same years as (a–c) and each divided by its area mean *i.e.* such that all values would be unity for a uniform field. Vertical dashed lines delimit the 30th percentile bounds as used in the SST# measure, whilst horizontal dashed lines denote average values within the temperature ranges that they span.

distinguished from unforced variability. The Krakatoa and Pinatubo eruptions are qualitatively similar in terms of radiative forcing, as both were equatorial and began in NH summer months (Mass & Portman, 1989; Stenchikov et al., 1998).

For other forcings, we consider 1995–2015. Figure 10 shows time-mean forcing and temperature change for all the forcings, computed from time means each divided by their signed area means; hence they are mostly positive and the area-mean is +1 in all cases, including negative hist-aer and volcanic forcing. Figures 10a and 10b show zonal-mean forcing at the TOA and at the surface, evaluated as described in Section 2.2. Volcanic forcing does not exhibit any unusual feature as a function of latitude that could account for the low climate sensitivity, in contrast to hist-aer, which has strong tropical-extratropical and NH-SH contrasts. The differences among forcings are reflected in the zonal mean surface temperature changes (Figure 10c). Temperature change is greater at higher northern latitudes (around 40N) for hist-aer compared to hist-GHG, whilst for volcanic forcing, which might be due to Antarctic sea ice change. This may stem from the negative SH forcing under volcanic eruptions, which does not occur in the other cases, with the sea ice response being larger under cooling than under warming.

Low climate sensitivity (large α) is correlated with large dS/dT (Equation 5), which results from enhanced surface warming in tropical regions of deep convection, especially the West Pacific warm pool (Andrews & Webb, 2018; Ceppi & Gregory, 2017; Fueglistaler, 2019), where climatological SST is highest. In these regions, warming propagates easily from the surface to the atmosphere. Warming in less convective regions, by contrast, causes little change in free-tropospheric temperature. Thus the difference between warming in the warm pool and elsewhere is indicative of the change in tropospheric stability. To focus on this effect, we evaluate the distribution of TOA forcing, surface forcing, and temperature change over the sea in the tropics (30°S–30°N) as a function of the area-weighted distribution of tropical climatological SST (Figures 10d and 10e). In these distributions, equal intervals in SST percentile (horizontal axis) correspond to equal surface areas, so that a field which is uniform in space would have a constant value as a function of SST percentile. Following Fueglistaler (2019), we define the warmest 30% of tropical SSTs as the "warm pool."



At the TOA (Figure 10d), in contrast to other historical forcings, volcanic forcing is greater over the warm pool than in the rest of the tropics. This might explain why SST change (Figure 10f) in the warm pool is greater than elsewhere, which could then explain the significantly greater dS/dT for volcanic forcing. Such enhanced warm pool SST temperature change has been found in other work when looking at Pinatubo-style eruptions (Günther et al., 2022). Evidence against this argument is that volcanic forcing *at the surface* does not seem to be significantly larger in the warm pool than in the rest of the tropics when contrasted with other historical forcings (Figure 10e), while we expect surface temperature change to be more related to surface than TOA forcing.

We propose another mechanism that may contribute to the significantly larger *dS/dT* and hence smaller temperature change for volcanic forcing than for GHG forcing. The response of the upper ocean mainly determines the SST change to volcanic forcing, because it appears and disappears in only a few years, too quickly for the slower processes of heat exchange with the deeper ocean (Gregory et al., 2015, 2016; Held et al., 2010). This is less relevant for historical AA and GHG forcings, which evolve slowly over time. Therefore the geographical variation of upper-ocean heat capacity or wind-driven advection might dominate the pattern of SST change for fairly uniform forcings over very short timescales. If this pattern correlates with climatological SSTs, stability changes per unit of temperature change may be larger.

Rugenstein et al. (2016) provide insight into what this pattern might look like. For abrupt $4 \times CO_2$, they find greater warming in the warm pool and less outside of deep convective regions in early years (see their Figure 4). This relatively larger warm pool warming disappears after a few years, suggesting a strong SST pattern–driven stability response initially that becomes weaker over time, per unit of warming. If this is the characteristic pattern of the response to short-lived forcing, it could explain why volcanic events cause greater dS/dT (and α) than other historical forcing agents.

8. Summary and Conclusions

The findings of this work can be summarized as follows.

- In the MMM of CMIP6 historical experiments, the time-variation of the radiative feedback parameter α (a positive number, larger for smaller effective climate sensitivity) is substantial, and driven by volcanic forcing and the changing ratio of AA and greenhouse-gas (GHG) forcing. Volcanic forcing dominates before about 1915, then AA and GHG up to about 1960. All three kinds of forcings contribute in more recent years.
- We confirm that the dependence of radiative feedback on forcing agent can be explained through differences in *dS/dT*, the sensitivity of global-mean tropospheric stability to global-mean surface temperature (Ceppi & Gregory, 2019; Salvi et al., 2022).
- Time-variation of dS/dT explains the forced time-variation in α for each forcing agent individually, and for all agents together, as long as the forcing signal is sufficiently large to cause variations in dS/dT and α that exceed unforced variability.
- Time-variation of *dS/dT* in the MMM of amip-piForcing experiments (with observed SST and no radiative forcing) explains the time-variation of *α* in this case, which differs from the CMIP6 MMM historical *α*, possibly because AOGCMs fail to recreate observed SST patterns (especially in recent decades).
- Shortwave (SW) cloud feedbacks play the largest role for net feedback differences among forcing agents. Non-cloud feedbacks (SW and longwave combined) contribute significantly but to a lesser extent to the differences. This augments the body of research on the SST pattern effect (Stevens et al., 2016) being related to radiative feedbacks through stability and SW cloud feedbacks (Ceppi & Gregory, 2017, 2019; Kamae et al., 2019; Mackie et al., 2021).
- Analysis of the strongest two eruptions of the 1850–2015 period (Pinatubo and Krakatoa) highlight very large *dS/dT* for volcanic eruptions, which results in lower climate sensitivity from their forcing.
- Large *dS/dT* for volcanic eruptions is tied to strong SST response (cooling) in the west Pacific warm pool. This might be linked to volcanic forcing being stronger over the warm pool, or it might be characteristic of the rapid response of the ocean to forcing.

Our findings are broadly consistent with those of Gregory et al. (2020), who analyzed CMIP5 data. We reaffirm that α for GHG forcing has little time-variation, but that there is significant time-variation in α for the combination of historical forcings, which is to some extent a result of the different time-evolution of the magnitudes of the various forcings, at least in the MMM.



21698996, 2023, 20, Downloaded from https://agupubs

wiley.com/doi/10.1029/2023JD038984 by Test, Wiley Online Library on [20/12/2023]. See the Terms and Conditions (https://wiley.com/doi/10.1029/2023JD038984 by Test, Wiley Online Library on [20/12/2023].

Unlike Gregory et al. (2020), we find significant time-variation of α early in hist-nat, maybe because we have knowledge of evolving effective forcing from the CMIP6 piClim experiments with individual models, while they had to estimate it from aerosol optical depth, which was prescribed in all models. Knowledge of the strong forcing from Pinatubo (1991) also leads to a different explanation of the final years of the historical simulation, when Gregory et al. (2020) posited that GHG forcing dominated. We suggest that α being less positive (larger effective climate sensitivity) for AA than for GHG, and more positive for volcanic forcing than for GHG, together leave historical α similar to GHG α in these years.

Our results are mostly consistent with Dong et al. (2021), who also used CMIP6 data, though we find more significant time-variation in feedbacks for the historical experiment, especially during the period of slowly varying Taround 1940. This is likely related to our use of regressions on the MMM as opposed to Dong et al. taking the mean of model regressions; we expect their method to result in regression dilution (Gregory et al., 2020), which causes the magnitude of the slope α to be underestimated. We do not expect the difference to be due to their having included data from GFDL-CM4, which we did not.

Other works have studied the evolution of feedbacks following an abrupt change in CO₂ forcing, and how this is related to the pattern effect (Andrews et al., 2012, 2015; Paynter et al., 2018; Rohrschneider et al., 2019; Rugenstein et al., 2020). It may be helpful to compare these abrupt CO₂ forcing cases to cases involving abrupt forcings from other agents, such as anthropogenic or volcanic aerosols. Experiments could also be performed with prescribed step TOA forcings in various patterns, in order to identify how climatological SSTs relate to SST changes under different forcing patterns with the same forcing timescales. Such work would give information about whether differences in forcing timescale are the main driver of differences in dS/dT between GHG and volcanic forcings, by disentangling the contributions of forcing patterns and forcing timescales from SST pattern changes.

The small number of large-magnitude volcanic forcing events in the historical period leaves it unclear whether such eruptions would generally cause high dS/dT, rather than it happening by chance in the cases we have studied. Further insight could be gained from experiments with synthetic data sets of prescribed stratospheric aerosol to mimic volcanic eruptions, perhaps at different latitudes or times of the year (as done in Zhuo et al. (2021)), or by simulating stronger versions, or larger ensembles, of historically weaker volcanic eruptions. The Model Intercomparison Project on the climatic response to Volcanic forcing (VolMIP; Zanchettin et al., 2016) aims to address many aspects of uncertainty with regards to volcanic forcings, but runs of volc-long-eq (long-term response to volcanic forcing) are only available for four models and there is not a concurrent effective forcing data set.

Experiments with a complete set of individual historical forcing agents would also be useful, including volcanic and solar separately (e.g., hist-volc and hist-sol; Gillett et al., 2016), land use, ozone, black carbon and sulphate aerosols. For each agent, historical AOGCM and AGCM experiments are needed, the latter with climatological SST in order to get the time-evolving effective radiative forcing required to calculate feedbacks. These further experiments would help complete the explanation of the historical pattern effect on radiative feedback.

Data Availability Statement

The data used here, from CMIP6 experiments (Eyring et al., 2016), can be found on the Earth System Grid Federation (Cinquini et al., 2014) CEDA node available at esgf-index1.ceda.ac.uk.

References

Ammann, C. M., Meehl, G. A., Washington, W. M., & Zender, C. S. (2003). A monthly and latitudinally varying volcanic forcing dataset in simulations of 20th century climate. Geophysical Research Letters, 30(12). https://doi.org/10.1029/2003GL016875

Andrews, T., Bodas-Salcedo, A., Gregory, J. M., Dong, Y., Armour, K. C., Paynter, D., et al. (2022). On the effect of historical SST patterns on radiative feedback. Journal of Geophysical Research: Atmospheres, 127(18), e2022JD036675. https://doi.org/10.1029/2022JD036675

Andrews, T., Gregory, J. M., Paynter, D., Silvers, L. G., Zhou, C., Mauritsen, T., et al. (2018). Accounting for changing temperature patterns increases historical estimates of climate sensitivity. Geophysical Research Letters, 45(16), 8490-8499. https://doi.org/10.1029/2018GL078887 Andrews, T., Gregory, J. M., & Webb, M. J. (2015). The dependence of radiative forcing and feedback on evolving patterns of surface temperature change in climate models. Journal of Climate, 28(4), 1630-1648. https://doi.org/10.1175/JCLI-D-14-00545.1

Andrews, T., & Webb, M. J. (2018). The dependence of global cloud and lapse rate feedbacks on the spatial structure of tropical Pacific warming. Journal of Climate, 31(2), 641-654. https://doi.org/10.1175/JCLI-D-17-0087.1

Acknowledgments

We thank two reviewers (Maria Rugenstein and anonymous) for helpful and constructive comments. PS was supported by the Department of Physics and the Grantham Institute of Imperial College London, as well as the European Research Council under the European Union's Horizon 2020 research and innovation programme (grant agreement 786427, project "Couplet"). PC was supported by an Imperial College Research Fellowship and NERC grants NE/T006250/1, NE/T007788/1, and NE/ V012045/1. JMG was supported by the European Research Council under the European Union's Horizon 2020 research and innovation programme (Grant agreement 786427, project "Couplet"). This work used JASMIN, the UK collaborative data analysis facility.

terms-and-conditions) on Wiley Online Library for rules of use; OA

articles

are governed by the applicable Creative Commons Licens

Andrews, T., Gregory, J. M., Webb, M. J., & Taylor, K. E. (2012). Forcing, feedbacks and climate sensitivity in CMIP5 coupled atmosphere-ocean climate models. Geophysical Research Letters, 39(9). https://doi.org/10.1029/2012GL051607

- Arfeuille, F., Weisenstein, D., Mack, H., Rozanov, E., Peter, T., & Brönnimann, S. (2014). Volcanic forcing for climate modeling: A new microphysics-based data set covering years 1600–present. *Climate of the Past*, 10(1), 359–375. https://doi.org/10.5194/cp-10-359-2014
- Armour, K. C., Bitz, C. M., & Roe, G. H. (2013). Time-varying climate sensitivity from regional feedbacks. Journal of Climate, 26(13), 4518– 4534. https://doi.org/10.1175/JCLI-D-12-00544.1
- Boggs, P. T., & Rogers, J. E. (1990). Orthogonal distance regression (Tech. Rep.). Center for computing and applied mathematics. Retrieved from https://docs.scipy.org/doc/external/odr_ams.pdf
- Ceppi, P., Brient, F., Zelinka, M. D., & Hartmann, D. L. (2017). Cloud feedback mechanisms and their representation in global climate models. Wiley Interdisciplinary Reviews: Climate Change, 8(4), e465. https://doi.org/10.1002/wcc.465
- Ceppi, P., & Gregory, J. M. (2017). Relationship of tropospheric stability to climate sensitivity and Earth's observed radiation budget. Proceedings of the National Academy of Sciences, 114(50), 13126–13131. https://doi.org/10.1073/pnas.1714308114
- Ceppi, P., & Gregory, J. M. (2019). A refined model for the Earth's global energy balance. *Climate Dynamics*, 53(7), 4781–4797. https://doi.org/10.1007/s00382-019-04825-x
- Cinquini, L., Crichton, D., Mattmann, C., Harney, J., Shipman, G., Wang, F., et al. (2014). The Earth System Grid Federation: An open infrastructure for access to distributed geospatial data. *Future Generation Computer Systems*, 36, 400–417. https://doi.org/10.1016/j.future.2013.07.002
- Dong, Y., Armour, K. C., Proistosescu, C., Andrews, T., Battisti, D. S., Forster, P. M., et al. (2021). Biased estimates of equilibrium climate sensitivity and transient climate response derived from historical CMIP6 simulations. *Geophysical Research Letters*, 48(24), e2021GL095778. https://doi.org/10.1029/2021GL095778
- Dong, Y., Armour, K. C., Zelinka, M. D., Proistosescu, C., Battisti, D. S., Zhou, C., & Andrews, T. (2020). Intermodel spread in the pattern effect and its contribution to climate sensitivity in CMIP5 and CMIP6 models. *Journal of Climate*, 33(18), 7755–7775. https://doi.org/10.1175/ JCLI-D-19-1011.1
- Dong, Y., Proistosescu, C., Armour, K. C., & Battisti, D. S. (2019). Attributing historical and future evolution of radiative feedbacks to regional warming patterns using a Green's function approach: The Preeminence of the western Pacific. *Journal of Climate*, 32(17), 5471–5491. https:// doi.org/10.1175/JCLI-D-18-0843.1
- Durack, P. J., & Taylor, K. E. (2022). CMIP6 forcing data sets summary v6.2.43. Retrieved from http://goo.gl/r8up31
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Fueglistaler, S. (2019). Observational evidence for two modes of coupling between sea surface temperatures, tropospheric temperature Profile, and shortwave cloud radiative effect in the tropics. *Geophysical Research Letters*, 46(16), 9890–9898. https://doi.org/10.1029/2019GL083990
- Fueglistaler, S., & Silvers, L. G. (2021). The peculiar trajectory of global warming. *Journal of Geophysical Research: Atmospheres*, 126(4), e2020JD033629. https://doi.org/10.1029/2020JD033629
- Gillett, N. P., Shiogama, H., Funke, B., Hegerl, G., Knutti, R., Matthes, K., et al. (2016). Detection and Attribution Model Intercomparison Project (DAMIP) (preprint). Climate and Earth System Modeling. https://doi.org/10.5194/gmd-2016-74
- Gregory, J. M., & Andrews, T. (2016). Variation in climate sensitivity and feedback parameters during the historical period. *Geophysical Research Letters*, 43(8), 3911–3920. https://doi.org/10.1002/2016GL068406
- Gregory, J. M., Andrews, T., Ceppi, P., Mauritsen, T., & Webb, M. J. (2020). How accurately can the climate sensitivity to CO2 be estimated from historical climate change? *Climate Dynamics*, 54(1), 129–157. https://doi.org/10.1007/s00382-019-04991-y
- Gregory, J. M., Andrews, T., & Good, P. (2015). Inconstancy of the transient climate response parameter under increasing CO₂. Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences, 373(2054), 20140417. https://doi.org/10.1098/ rsta.2014.0417
- Gregory, J. M., Andrews, T., Good, P., Mauritsen, T., & Forster, P. M. (2016). Small global-mean cooling due to volcanic radiative forcing. *Climate Dynamics*, 47(12), 3979–3991. https://doi.org/10.1007/s00382-016-3055-1
- Gregory, J. M., Bi, D., Collier, M., Dix, M., Hirst, A., Hu, A., et al. (2013). Climate models without preindustrial volcanic forcing underestimate historical ocean thermal expansion. *Geophysical Research Letters*, 40(8), 1600–1604. https://doi.org/10.1002/grl.50339
- Gregory, J. M., Stouffer, R. J., Raper, S. C. B., Stott, P. A., & Rayner, N. A. (2002). An observationally based estimate of the climate sensitivity. Journal of Climate, 15(22), 3117–3121. https://doi.org/10.1175/1520-0442(2002)015(3117:AOBEOT)2.0.CO;2
- Günther, M., Schmidt, H., Timmreck, C., & Toohey, M. (2022). Climate feedback to stratospheric aerosol forcing: The key role of the pattern effect. Journal of Climate, 35(24), 7903–7917. https://doi.org/10.1175/JCLI-D-22-0306.1
- Held, I. M., Winton, M., Takahashi, K., Delworth, T., Zeng, F., & Vallis, G. K. (2010). Probing the fast and slow components of global warming by returning abruptly to preindustrial forcing. *Journal of Climate*, 23(9), 2418–2427. https://doi.org/10.1175/2009JCL13466.1
- Hurrell, J. W., Hack, J. J., Shea, D., Caron, J. M., & Rosinski, J. (2008). A new sea surface temperature and sea ice boundary data set for the community atmosphere model. *Journal of Climate*, 21(19), 5145–5153. https://doi.org/10.1175/2008JCLI2292.1
- Jeevanjee, N., Koll, D. D. B., & Lutsko, N. (2021). Simpson's Law" and the spectral cancellation of climate feedbacks. *Geophysical Research Letters*, 48(14), e2021GL093699. https://doi.org/10.1029/2021GL093699
- Kamae, Y., Chadwick, R., Ackerley, D., Ringer, M., & Ogura, T. (2019). Seasonally variant low cloud adjustment over cool oceans. *Climate Dynamics*, 52(9–10), 5801–5817. https://doi.org/10.1007/s00382-018-4478-7
- Lamarque, J.-F., Bond, T. C., Eyring, V., Granier, C., Heil, A., Klimont, Z., et al. (2010). Historical (1850–2000) gridded anthropogenic and biomass burning emissions of reactive gases and aerosols: Methodology and application. *Atmospheric Chemistry and Physics*, 10(15), 7017– 7039. https://doi.org/10.5194/acp-10-7017-2010
- Mackie, A., Brindley, H. E., & Palmer, P. I. (2021). Contrasting observed atmospheric responses to tropical sea surface temperature warming patterns. Journal of Geophysical Research: Atmospheres, 126(7), e2020JD033564. https://doi.org/10.1029/2020JD033564
- Marvel, K., Schmidt, G. A., Miller, R. L., & Nazarenko, L. S. (2016). Implications for climate sensitivity from the response to individual forcings. *Nature Climate Change*, 6(4), 386–389. https://doi.org/10.1038/nclimate2888
- Mass, C. F., & Portman, D. A. (1989). Major volcanic eruptions and climate: A critical evaluation. Journal of Climate, 2(6), 566–593. https://doi. org/10.1175/1520-0442(1989)002(0566:MVEACA)2.0.CO;2
- Myhre, G., Shindell, D., Bréon, F.-M., Collins, W., Fuglestvedt, J., Huang, J., et al. (2014). Anthropogenic and natural radiative forcing. (Tech. Rep. No. AR5). In Climate Change 2013: The Physical Science Basis. Contribution of Working Group 1 to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press. Retrieved from https://www.ipcc.ch/report/ar5/wg1/
- Newhall, C. G., & Self, S. (1982). The volcanic explosivity index (VEI) an estimate of explosive magnitude for historical volcanism. Journal of Geophysical Research, 87(C2), 1231–1238. https://doi.org/10.1029/JC087iC02p01231



- Nordling, K., Korhonen, H., Räisänen, J., Partanen, A.-I., Samset, B. H., & Merikanto, J. (2021). Understanding the surface temperature response and its uncertainty to CO₂, CH₄, black carbon, and sulfate. *Atmospheric Chemistry and Physics*, 21(19), 14941–14958. https://doi.org/10.5194/ acp-21-14941-2021
- Paynter, D., Frölicher, T. L., Horowitz, L. W., & Silvers, L. G. (2018). Equilibrium climate sensitivity obtained from multimillennial runs of two GFDL climate models. *Journal of Geophysical Research: Atmospheres*, 123(4), 1921–1941. https://doi.org/10.1002/2017JD027885

Pendergrass, A. G., Conley, A., & Vitt, F. M. (2018). Surface and top-of-atmosphere radiative feedback kernels for CESM-CAM5. Earth System Science Data, 10(1), 317–324. https://doi.org/10.5194/essd-10-317-2018

- Persad, G. G., & Caldeira, K. (2018). Divergent global-scale temperature effects from identical aerosols emitted in different regions. *Nature Communications*, 9(1), 3289. https://doi.org/10.1038/s41467-018-05838-6
- Picas, J., & Grab, S. (2020). Potential impacts of major nineteenth century volcanic eruptions on temperature over cape town, South Africa: 1834–1899. Climatic Change, 159(4), 523–544. https://doi.org/10.1007/s10584-020-02678-6
- Pincus, R., Forster, P. M., & Stevens, B. (2016). The radiative forcing model Intercomparison project (RFMIP): Experimental protocol for CMIP6. Geoscientific Model Development, 9(9), 3447–3460. https://doi.org/10.5194/gmd-9-3447-2016
- Proistosescu, C., & Huybers, P. J. (2017). Slow climate mode reconciles historical and model-based estimates of climate sensitivity. *Science Advances*, 3(7), e1602821. https://doi.org/10.1126/sciadv.1602821
- Richardson, T. B., Forster, P. M., Smith, C. J., Maycock, A. C., Wood, T., Andrews, T., et al. (2019). Efficacy of climate forcings in PDRMIP models. Journal of Geophysical Research: Atmospheres, 124(23), 12824–12844. https://doi.org/10.1029/2019JD030581
- Rohrschneider, T., Stevens, B., & Mauritsen, T. (2019). On simple representations of the climate response to external radiative forcing. *Climate Dynamics*, 53(5), 3131–3145. https://doi.org/10.1007/s00382-019-04686-4
- Rugenstein, M., & Armour, K. C. (2021). Three flavors of radiative feedbacks and their implications for estimating equilibrium climate sensitivity. Geophysical Research Letters, 48(15), e2021GL092983. https://doi.org/10.1029/2021GL092983
- Rugenstein, M., Bloch-Johnson, J., Abe-Ouchi, A., Andrews, T., Beyerle, U., Cao, L., et al. (2019). LongRunMIP: Motivation and design for a large collection of millennial-Length AOGCM simulations. *Bulletin of the American Meteorological Society*, 100(12), 2551–2570. American Meteorological Society Section. https://doi.org/10.1175/BAMS-D-19-0068.1
- Rugenstein, M., Bloch-Johnson, J., Gregory, J., Andrews, T., Mauritsen, T., Li, C., et al. (2020). Equilibrium climate sensitivity estimated by equilibrating climate models. *Geophysical Research Letters*, 47(4), e2019GL083898. https://doi.org/10.1029/2019GL083898
- Rugenstein, M., Gregory, J. M., Schaller, N., Sedláček, J., & Knutti, R. (2016). Multiannual ocean–atmosphere adjustments to radiative forcing. *Journal of Climate*, 29(15), 5643–5659. https://doi.org/10.1175/JCLI-D-16-0312.1
- Salvi, P., Ceppi, P., & Gregory, J. M. (2022). Interpreting differences in radiative feedbacks from aerosols versus greenhouse gases. *Geophysical Research Letters*, 49(8), e2022GL097766. https://doi.org/10.1029/2022GL097766
- Schmidt, A., Mills, M. J., Ghan, S., Gregory, J. M., Allan, R. P., Andrews, T., et al. (2018). Volcanic radiative forcing from 1979 to 2015. Journal of Geophysical Research: Atmospheres, 123(22), 12491–12508. https://doi.org/10.1029/2018JD028776
- Shindell, D. T. (2014). Inhomogeneous forcing and transient climate sensitivity. *Nature Climate Change*, 4(4), 274–277. https://doi.org/10.1038/ nclimate2136
- Smith, C. J., & Forster, P. M. (2021). Suppressed late-20th century warming in CMIP6 models explained by forcing and feedbacks. *Geophysical Research Letters*, 48(19), e2021GL094948. https://doi.org/10.1029/2021GL094948
- Soden, B. J., Held, I. M., Colman, R., Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Quantifying climate feedbacks using radiative kernels. Journal of Climate, 21(14), 3504–3520. https://doi.org/10.1175/2007JCLI2110.1
- Stenchikov, G. L., Kirchner, I., Robock, A., Graf, H.-F., Antuña, J. C., Grainger, R. G., et al. (1998). Radiative forcing from the 1991 Mount Pinatubo volcanic eruption. *Journal of Geophysical Research*, 103(D12), 13837–13857. https://doi.org/10.1029/98JD00693
- Stevens, B., Sherwood, S. C., Bony, S., & Webb, M. J. (2016). Prospects for narrowing bounds on Earth's equilibrium climate sensitivity. Earth's Future, 4(11), 512–522. https://doi.org/10.1002/2016EF000376
- Taylor, K. E., Crucifix, M., Braconnot, P., Hewitt, C. D., Doutriaux, C., Broccoli, A. J., et al. (2007). Estimating shortwave radiative forcing and response in climate models. *Journal of Climate*, 20(11), 2530–2543. American Meteorological Society Section. https://doi.org/10.1175/ JCLI4143.1
- Wang, H., & Wen, Y.-J. (2021). Climate response to the spatial and temporal evolutions of anthropogenic aerosol forcing. *Climate Dynamics*, 59(5–6), 1579–1595. https://doi.org/10.1007/s00382-021-06059-2
- Webb, M. J., Andrews, T., Bodas-Salcedo, A., Bony, S., Bretherton, C. S., Chadwick, R., et al. (2017). The cloud feedback model Intercomparison project (CFMIP) contribution to CMIP6. *Geoscientific Model Development*, 10(1), 359–384. Copernicus GmbH. https://doi.org/10.5194/gmd-10-359-2017
- Wiart, P., & Oppenheimer, C. (2000). Largest known historical eruption in Africa: Dubbi volcano, Eritrea, 1861. Geology, 28(4), 291–294. https://doi.org/10.1130/0091-7613(2000)28(291:LKHEIA)2.0.CO;2
- Wills, R. C. J., Dong, Y., Proistosecu, C., Armour, K. C., & Battisti, D. S. (2022). Systematic climate model biases in the large-scale patterns of recent sea-surface temperature and sea-level Pressure change. *Geophysical Research Letters*, 49(17), e2022GL100011. https://doi. org/10.1029/2022GL100011
- Wood, R., & Bretherton, C. S. (2006). On the relationship between stratiform low cloud cover and lower-tropospheric stability. Journal of Climate, 19(24), 6425–6432. https://doi.org/10.1175/JCLI3988.1
- Zanchettin, D., Khodri, M., Timmreck, C., Toohey, M., Schmidt, A., Gerber, E. P., et al. (2016). The model Intercomparison project on the climatic response to volcanic forcing (VolMIP): Experimental design and forcing input data for CMIP6. *Geoscientific Model Development*, 9(8), 2701–2719. https://doi.org/10.5194/gmd-9-2701-2016
- Zelinka, M. D., Myers, T. A., McCoy, D. T., Po-Chedley, S., Caldwell, P. M., Ceppi, P., et al. (2020). Causes of higher climate sensitivity in CMIP6 models. *Geophysical Research Letters*, 47(1), e2019GL085782. https://doi.org/10.1029/2019GL085782
- Zhou, C., Zelinka, M. D., & Klein, S. A. (2016a). Impact of decadal cloud variations on the Earth's energy budget. Nature Geoscience, 9(12), 871–874. https://doi.org/10.1038/ngeo2828
- Zhou, C., Zelinka, M. D., & Klein, S. A. (2017). Analyzing the dependence of global cloud feedback on the spatial pattern of sea surface temperature change with a Green's function approach. *Journal of Advances in Modeling Earth Systems*, 9(5), 2174–2189. https://doi. org/10.1002/2017MS001096
- Zhou, T., Turner, A. G., Kinter, J. L., Wang, B., Qian, Y., Chen, X., et al. (2016b). GMMIP (v1.0) contribution to CMIP6: Global monsoons model inter-comparison project. *Geoscientific Model Development*, 9(10), 3589–3604. https://doi.org/10.5194/gmd-9-3589-2016
- Zhuo, Z., Kirchner, I., Pfahl, S., & Cubasch, U. (2021). Impact of volcanic eruptions: The sensitivity to eruption season and latitude in MPI-ESM ensemble experiments. Atmospheric Chemistry and Physics, 21(17), 13425–13442. https://doi.org/10.5194/acp-21-13425-2021